Author’s response to reviews

Title: Application of GRADE: Making Evidence-Based Recommendations about Diagnostic Tests in Clinical Practice Guidelines

Authors:

Jonathan Hsu (hsuj@mcmaster.ca)
Jan L Brozek (brozekj@mcmaster.ca)
Luigi Terraciano (terrycom1957@gmail.com)
Julia Kreis (j.kreis@iqwig.de)
Enrico Compalati (enrico.compalati@gmail.com)
Airton Tetelboom Stein (astein@via-rs.net)
Alessandro Fiocchi (alexfioc@gmail.com)
Holger J Schünemann (schuneh@mcmaster.ca)

Version: 3 Date: 17 May 2011

Author’s response to reviews:

To whom it may concern,

This version includes all the copy edit changes that were requested. No other changes have been made and all suggestions were accepted.

Best,

Holger Schunemann